

Dear editors,

I am glad to receive the valuable comments and suggestions from the three editors. The comments made by each editor were individually revised and highlighted in yellow in the article.

First of all, the first editor: 1. For abbreviations, we have made a list of abbreviations that we want to relate to. 2. For "NAFLD", we have replaced it with "MAFLD" in its entirety. 3. We have made changes to the full-text format. 4. For the schematic signal diagram you mentioned, we have also added the signal diagram in the summary section according to the results of the article.

Secondly, the second editor: 1. We have also made changes to the introduction, conclusion, language and logic of the discussion. 2. We have also defined the P-value. 3. At the same time, we have added an "Ethics Review Form" to the attached information.

And finally, the third editor: Thank you very much for your recognition of our article, thank you.

These are our responses to the questions posed by the three editors, and thank you very much for your valuable suggestions. Have a great day.

Best regards,

Wenmin Dong